Author's response to reviews

Title: The Effect of Incident Tuberculosis on Immunological Response of HIV Patients on Highly Active Antiretroviral Therapy at Gondar University Hospital, Northwest Ethiopia: A Retrospective Follow-up Study.

Authors:

Abate Assefa (abezew@gmail.com)
Baye Gelaw (tedybayegelaw@gmail.com)
Gebeyaw Getnet (ggebeyaw@yahoo.com)
Gashaw Yitayew (gashawyitayew@yahoo.com)

Version: 2 Date: 1 March 2014

Author's response to reviews:

Dear editor,

The legends of the figures are included accordingly in the main part of the manuscript next to the references.

Best regards!